Author's response to reviews

Title: What Motivates Senior Clinicians To Teach Medical Students?

Authors:

Jane Dahlstrom (jane.dahlstrom@act.gov.au)
Anna Dorai-Raj (dr_anna_dorai_raj@hotmail.com)
Darryl McGill (darryl.mcgill@act.gov.au)
Cathy Owen (cathy.owen@anu.edu.au)
Kathleen Tymms (canberra_rheumatology@bigpond.com)
Ashley Watson (Ashley.Watson@act.gov.au)

Version: 2 Date: 24 March 2005

Author's response to reviews:

Dear editors

sorry realised a number typo and fixed it

CHeers

Prof Cathy Owen